Author's response to reviews

Title: Interventions aimed at improving the nursing work environment: a systematic review

Authors:

Donna MJ Schalk (donna.schalk@han.nl)
Marloes LP Bijl (m.bijl@student.maastrichtuniversity.nl)
Greta G Cummings (gretac@ualberta.ca)
Ruud JG Halfens (r.halfens@zw.unimaas.nl)
Louk Hollands (l.hollands@zw.unimaas.nl)

Version: 8 Date: 2 April 2010

Author's response to reviews:

Dear Penny

thank you for the excellent copy editing. I have gone through your edits and accepted all but two. On page 9, line 6, and one page 13, line 2, NWE had been replaced by new. I have changed them both back to NWE to represent the Nursing Work Environment.

The font changes to size 12 resulted in line spacing difficulties in Table 4. I have rectified those to ensure that information across columns are appropriately lined up. I also replaced hyphens in Table 4 with Endashes to make the negative sign more legible. All changes have been accepted in the resubmitted manuscript.

We have double checked the supplemental documents and only have a question. Do we need to put a header or footer on those documents to reflect their connection to this manuscript when someone downloads them? Or will this information be inserted automatically?

Thank you and best wishes

greta